Author's response to reviews

Title: Geographical variation in cardiovascular incidence: results from the British Women's Heart and Health Study

Authors:

Lois G Kim (lois.kim@lshtm.ac.uk)
Claire Carson (claire.carson@npeu.ox.ac.uk)
Debbie A Lawlor (d.a.lawlor@bristol.ac.uk)
Shah Ebrahim (shah@ccdcindia.org)

Version: 5 Date: 11 November 2010

Author's response to reviews:

Additional files are now referred to in the text.